Author's response to reviews

**Title:** Development of lower limb range of motion from early childhood to adolescence in cerebral palsy - a population-based study.

**Authors:**

Eva Nordmark (eva.nordmark@med.lu.se)  
Gunnar Hägglund (gunnar.hagglund@med.lu.se)  
Henrik Lauge-Pedersen (henrik.lauge-pedersen@med.lu.se)  
Philippe Wagner (pw@nko.se)  
Lena Westbom (lena.westbom@med.lu.se)

**Version:** 2  **Date:** 11 September 2009

**Author's response to reviews:**

Dear reviewers and editors, we would like to thank the reviewers for valuable advice and are happy to publish our article in BMC Medicine. We are now completing the submission. According to the payment, Lund University became a member of BioMed Central in February 2002. Best regards Eva Nordmark